Author's response to reviews

Title: An exploratory study of the socio-cultural risk influences for cigarette smoking among Southern Nigerian youth

Authors:

Catherine O Egbe (egbe@ukzn.ac.za)
Inge Petersen (peterseni@ukzn.ac.za)
Anna Meyer-Weitz (meyerweitz@ukzn.ac.za)
Kwaku Oppong Asante (oppongasante@ukzn.ac.za)

Version: 2 Date: 2 October 2014

Author's response to reviews:

Dear Editor,

We have made revisions to our manuscript as recommended by each reviewer. Please find all changes made and responses to each reviewer's recommendation in the Table of corrections attached.

We look forward to hearing from you.

With many thanks,
Catherine O. Egbe
(Corresponding Author)